### Check for updates

#### **OPEN ACCESS**

EDITED BY Ranjit Kumar Upadhyay, Indian Institute of Technology Dhanbad, India

REVIEWED BY Eric Brako Dompreh, WorldFish, Malaysia Furqan Asif, Aalborg University, Denmark

\*CORRESPONDENCE Ekemini M. Okon ⊠ okon.ekeminimoses@gmail.com

### <sup>†</sup>PRESENT ADDRESSES

Reuben C. Okocha, Landmark University SDG 13 (Climate Action Research Group), Omu-Aran, Nigeria Babatunde M. Falana, Landmark University SDG 14 (Life Below Water Research Group), Omu-Aran, Nigeria

RECEIVED 03 July 2023 ACCEPTED 22 April 2024 PUBLISHED 10 July 2024

#### CITATION

Okon EM, Okocha RC, Folowosele DO, Oluwafemi PT and Falana BM (2024) Aquatic food resources in tropical Africa: a comprehensive analysis. *Front. Sustain. Food Syst.* 8:1252119. doi: 10.3389/fsufs.2024.1252119

#### COPYRIGHT

© 2024 Okon, Okocha, Folowosele, Oluwafemi and Falana. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# Aquatic food resources in tropical Africa: a comprehensive analysis

Ekemini M. Okon<sup>1,2</sup>\*, Reuben C. Okocha<sup>3†</sup>, Deborah O. Folowosele<sup>3</sup>, Precious T. Oluwafemi<sup>3</sup> and Babatunde M. Falana<sup>3†</sup>

<sup>1</sup>Department of Animal Sciences and Aquatic Ecology, Faculty of Bioscience Engineering, Ghent University, Ghent, Belgium, <sup>2</sup>Department of Biology, Vegetal Biology and Ecology, Universitat Autònoma de Barcelona, Barcelona, Spain, <sup>3</sup>Department of Agriculture, Landmark University, Omu-Aran, Kwara, Nigeria

Aquatic food production holds great promise as a viable solution to tackle the pressing issue of meeting the protein requirements of the African populace. It plays a crucial role in food security and economic growth, especially in tropical Africa. With a fast-expanding population, there is a growing need for aquatic food sources to provide inexpensive protein and essential nutrients. Despite Africa's vast aquatic food source supplies and the importance of aquatic food in healthy and sustainable diets, several challenges hinder its full exploitation in the tropical African environment. This study, therefore, aims to assess the existing state of aquatic food resources in Africa, examine the nutritional advantages associated with these foods, and identify the problems that hinder their maximal utilization. The findings highlight the differences in aquatic protein consumption across various countries, emphasizing the importance of considering local factors when formulating nutritional interventions and policies. The comparison between the supply of protein from aquatic sources and land animals across African nations revealed a consistent trend of lower availability of aquatic protein. Africa, in particular, had relatively lower per capita consumption values of animal protein. The continent's average quantity of land animal food significantly surpassed the average aquatic food supply. This suggests a predominant reliance on land animal sources in African countries to meet dietary protein needs. This finding holds significant importance for policymakers, nutritionists, and stakeholders, highlighting the need for targeted strategies to improve protein accessibility in African countries. Moreover, it calls for attention to address any existing imbalances between the utilization of land and aquatic protein sources in the region.

KEYWORDS

aquatic food, aquaculture, fisheries, fish consumption, sustainability, tropical Africa

### **1** Introduction

With a global population exceeding 8 billion, tropical Africa accommodates about 1.4 billion individuals (Worldometer, 2023). However, the region is confronted with high poverty levels, signifying the need for alternative strategies to fulfill the protein demands of its residents (Schönfeldt and Hall, 2012). Recent data obtained in early 2023 indicates that Africa persists in facing substantial difficulties, with undernourishment reaching 20.15% in 2013 (Abegaz, 2018). These statistics highlight the pressing need for sustainable and inventive measures to tackle the nutritional necessities of the African populace and enhance their general welfare.

The United Nations' 2021 State of Food Security and Nutrition shows that about 282 million people lack regular access to sufficient, safe, and nourishing food to meet their dietary needs and sustain a healthy lifestyle. This scenario highlights the challenge of undernourishment in tropical Africa, influenced by several factors, such as constrained agricultural productivity, pervasive poverty and inequality, challenging climate and weather conditions, ongoing political instability and conflict, and inadequate infrastructure (Gashu et al., 2019). Given these complexities, exploring alternative sources of nutrition, such as aquatic foods, becomes imperative to address food insecurity in these regions.

Aquatic foods are plants and animals sourced from aquatic environments, including fish, shellfish, seaweed, and aquatic plants, for human consumption. There are abundant water resources in Africa, and aquatic food production presents a promising and feasible solution to address the urgent challenge of fulfilling the protein demands of the African populace (Chan et al., 2019; Belton et al., 2021). This solution assumes even greater significance when considering that more than 30% of the African population relies on fish as their primary source of animal protein [Food and Agriculture Organization (FAO), 2022b] and contributes over 20% of animal protein intake (Hollinger and Staatz, 2015; Chan et al., 2019). This highlights that the importance of fish, comprising both finfish and shellfish, cannot be overemphasized, as they serve as pivotal sources of nutrition (Chan et al., 2021). It further contributes 17% of animal protein and 7% of all proteins, thereby playing a vital role in providing essential nourishment to over three billion individuals in developing countries (Boyd et al., 2022). As the demand for aquatic foods, particularly fish, continues to rise, sustainable management practices are paramount to ensuring the long-term viability of aquaculture and fisheries and the livelihoods they support.

The African governments, regional organizations, and local communities are investing in sustainable fish production systems to meet the increasing demand and reduce dependence on imported fish (Naylor et al., 2021). Moreover, Africa possesses a vast untapped potential for the advancement of aquaculture. The continent's abundant freshwater resources and favorable climatic conditions offer an ideal environment for fish cultivation (Adeleke et al., 2020). Aquaculture can generate more avenues for income in rural communities, alleviate strain on wild fish populations, and enhance overall food production (Gephart et al., 2020). Therefore, prioritizing investments in research, development, infrastructure, and capacitybuilding initiatives can effectively harness the potential of aquaculture to meet the escalating demands for fish, reduce poverty and stimulate economic growth in African countries (Adeleke et al., 2020). However, realizing this full potential in Africa necessitates unified efforts and collaboration among governments, stakeholders, and international organizations (Jolly et al., 2023). Hence, there is a need to actively promote sustainable practices, enhance infrastructure, facilitate market access, and fortify institutional frameworks to strengthen the expansion of fish and other aquatic food production (Rahman, 2021).

Fish production in tropical Africa varies significantly across the countries and regions. Africa collectively produced 16.7 million metric tons of fish, with approximately 25% originating from aquaculture [Food and Agriculture Organization (FAO), 2018]. Egypt, Nigeria, and Uganda have emerged as the leading aquaculture producers in the continent, leveraging their geographical advantages, government backing, and technological advancements to excel in fish farming (Adeleke et al., 2020). Over the years, these countries have improved their aquaculture production potential to address food security, generate employment opportunities, and contribute to overall economic growth (Adeleke et al., 2020; Muringai et al., 2022). Specifically, Egypt has become a key player by substantially producing Nile tilapia and African catfish (Mansour et al., 2021). Similarly, Nigeria has made significant strides in catfish and tilapia farming (Kaleem and Sabi, 2021), while Uganda has positioned itself through its Nile perch cultivation (Cowx and Ogutu-Owhayo, 2019).

In other countries, Mauritania, Senegal, and Angola have established their prominence as notable traditional fisheries producers, capitalizing on their extensive coastlines and abundant marine resources (Deme et al., 2023). These countries possess a rich fishing heritage, where artisanal and small-scale fishing activities are vital in providing livelihoods for coastal communities. Their traditional fisheries sector has sustained local economies and preserved cultural practices while meeting the demands of both domestic and international markets with exceptional seafood of the highest quality (Deme et al., 2023; GlobeFish, 2023). However, Mozambique, Ghana, the Democratic Republic of Congo, Madagascar, and Tanzania are confronted with significant deficiencies in their fish production. These countries encounter a range of obstacles that impede their ability to meet the growing demand for fish (GlobeFish, 2023). Factors such as limited infrastructure, insufficient investment in the fisheries sector, ineffective governance, overfishing, and environmental degradation contribute to these deficits (Akegbejo-Samsons, 2022). For instance, Mozambique, situated along an extensive coastline and endowed with abundant marine resources, contends with illegal fishing practices and a lack of suitable management strategies (Zeller et al., 2021). Conversely, despite its long-standing fishing industry, Ghana faces challenges with depleted fish stocks caused by unsustainable fishing practices and inadequate regulatory frameworks (Okyere et al., 2023). Moreover, the Democratic Republic of Congo, Madagascar, and Tanzania encounter similar problems, including restricted access to modern fishing technologies, inadequate storage and processing facilities, and weak governance in fisheries management (Eyayu et al., 2023).

Despite the vast aquatic food source resources and the role of aquatic food in healthy and sustainable diets, several factors limit its full utilization in the tropical African context. Therefore, this study examines the current state of aquatic food resources in the region, analyses the nutritional benefits of aquatic food, the factors that could affect aquatic protein consumption, including their relationships, and identifies the barriers limiting their utilization. This study further explores the environmental implications of maximizing aquatic food utilization and the policy interventions in promoting sustainable aquatic food systems in tropical Africa. Finally, this study proposes strategies for promoting the utilization of aquatic food resources, thereby contributing to the development of sustainable and healthy diets for the population. This study is organized thus: next, Section 2 explains the methodology. Section 3 highlights the nutritional benefits and aquatic food consumption in tropical Africa. Then, section 4 discusses the barriers to adequate consumption and section 5 reports on sustainability and its health implications. Section 6 reports the recent aquatic policies for more accessible aquatic foods. Finally, sections 7 and 8 highlight the study's limitations and the conclusion.

TABLE 1 Distribution of aquatic species mostly consumed across African countries.

Country	Aquatic species consumption	References
Egypt, Nigeria, Ghana	River and marine fish such as catfish and mackerel,	Bradley et al. (2020), Food and
	herring, sardines	Agriculture Organization (FAO)
		(2021), and Hinrichsen et al. (2022)
South Africa	Prepared and preserved fish, molluscs, and crustaceans	Mbaka et al. (2022)
Benin, Burundi, Malawi, Rwanda, Uganda, Zambia	Tilapia	Food and Agriculture Organization
		(FAO) (2021)
Tanzania	Fish from the African Great Lakes region	Mbaka et al. (2022)
Congo, Botswana, Mauritius, Algeria, Central African Republic,	No specific information was found.	
Burkina Faso, Cameroon, Niger, Seychelles, Côte d'Ivoire, Democratic		
Republic of the Congo, Djibouti, Mali, Guinea, Guinea-Bissau, Lesotho,		
Liberia, Libya, Mauritania, Sao Tome and Principe, Sierra Leone, South		
Sudan, Togo, Angola, Cabo Verde, Chad, Comoros,		

## 2 Methodology

This review used the Web of Science, Scopus, and Google Scholar databases to extract data records on aquatic food resources in tropical Africa. The search period was set at "All years" to capture all possible articles with no language restriction. The search terms included "Aquaculture" OR "Fisheries" AND "Tropical Africa," "Aquatic Food Resources" AND "Tropical Africa," "Fish Farming" OR "Sustainable Fisheries" AND "Tropical Africa," "Fish Species" AND "Aquatic Biodiversity" AND "Tropical Africa," "Fish Species" AND "Aquatic conducted by reading all the retrieved documents and screening relevant studies. Relevant information related to the review topic was extracted to evaluate the existing information on this subject.

This study further explored the typical intricate relationship between aquatic protein consumption and critical factors such as population, annual gross domestic product (GDP), land protein intake, aquatic food imports, and aquatic food export. Data for population and GDP across the African countries were sourced from the World Bank,<sup>1</sup> while aquatic food import and export data were from the FAOSTAT statistical database.<sup>2</sup>

# 3 Nutritional benefits of aquatic food and consumption in tropical Africa

Studies have documented the nutritional benefits of aquatic food consumption, including proteins, essential amino acids, vitamins, minerals, and heart-healthy omega-3 fatty acids (Golden et al., 2021; Tacon, 2023). Their consumption is linked to numerous health benefits, including enhancing brain and eye health and providing a healthier alternative to red and processed meats, which can negatively impact well-being (Qian et al., 2020; Golden et al., 2021). However, the nutrient composition varies among aquatic species, primarily influenced by their fat content. For instance, carp species such as *Amblypharyngodon mola* and *Puntius sophore* contain high amounts of vitamin K, ranging between 402.9 and 786.9 IU per 100 g of flesh, highlighting their significance in human nutrition (Bianchi et al., 2022). Mackerel provides high-quality protein, vitamins D and B12,

omega-3 fatty acids (Ahern et al., 2021; Golden et al., 2021; Tacon, 2023). Shellfish such as mussels, clams, and oysters are rich in protein, iron, zinc, and vitamin B12 and are a good source of antioxidants and omega-3 fatty acids (Venugopal and Gopakumar, 2017; Arshad et al., 2022). The human body relies on dietary sources to obtain two crucial omega-3 fatty acids [eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA)], emphasizing their essentiality (Shahidi and Ambigaipalan, 2018). Furthermore, regular consumption of aquatic foods lowers the risk of stroke, depression, Alzheimer's disease, and other chronic conditions, supporting a healthy heart (Awuchi et al., 2022). Thus, aquatic foods, mainly fatty species like African Mackerel (*Scomber scombrus*) and Sardines (*Sardinella* spp.), are healthy options as they are abundant in these fatty acids (Calder, 2013; Calder, 2021).

minerals like selenium and phosphorus, and is a good source of

Furthermore, aquatic foods offer more than omega-3 fatty acids; they serve as an exceptional reservoir of high-quality protein, crucial for constructing and mending body tissues. They play a vital role in maintaining a robust immune system and facilitating the production of enzymes and hormones (Gutiérrez et al., 2019; Tacon et al., 2020). Moreover, they encompass all the essential amino acids the body cannot synthesize independently, making them excellent protein sources for plant-based alternatives. In contrast to landbased animals and plant-based substitutes, aquatic foods are especially rich sources of vitamin D and iodine (Craig et al., 2021). Additionally, the minimal levels of saturated fat in aquatic foods make them a healthier substitute for land-based animals, which often contain substantial amounts of saturated fat associated with an elevated risk of heart disease (Aldaya et al., 2021). Consequently, choosing sustainably sourced aquatic foods, particularly shellfish, is environmentally friendly and requires fewer resources and less energy for production than land-based animals (Ahern et al., 2021; Laso et al., 2022). Given Africa's diverse ecosystems and cultural practices, it is important to consider the regional diversity in consumption patterns.

The consumption patterns of African aquatic foods show considerable heterogeneity across the continent (Table 1). In West Africa, along the coastal countries, and across numerous small island states, fish can contribute about 60% or even more of the overall dietary protein intake (for instance, in countries like Gambia, Sierra Leone, and Ghana) [Food and Agriculture Organization (FAO), 2018]. While the overall average consumption of aquatic foods in Africa

<sup>1</sup> www.data.worldbank.org

<sup>2</sup> https://www.fao.org/faostat/en/#data/FBS

remains relatively low, it is noteworthy that eleven countries have surpassed the global average (Chan et al., 2021). These countries encompass some small island developing states (SIDS) and Gabon, Congo, the Gambia, Sierra Leone, Ghana, Egypt, and Côte d'Ivoire, indicating their relatively higher aquatic food consumption rates. Conversely, most African countries contend with relatively low levels of aquatic food consumption due to several factors, such as geographical, economic, and social factors (Rocha et al., 2022). However, despite these challenges, there is a growing recognition of the nutritional benefits and economic potential of increasing aquatic food consumption in many African nations.

The demand for fish has experienced a substantial and rapid surge in recent years (Iversen et al., 2020). This upward trajectory can be primarily attributed to the simultaneous growth in population and income levels (Oreggioni et al., 2021). Furthermore, increased awareness of the myriad health benefits of fish consumption has further propelled this trend (Thurstan and Roberts, 2014). Additionally, the ever-evolving dynamics of urbanization and globalization have significantly shaped consumer preferences and lifestyle choices, thus contributing to the increasing demand for fish. The per capita fish consumption in Africa is reported to gradually decrease from 10.0 kg/person/year (about half the global and Asian fish intake) in 2015 to7.7 kg/person/year in 2050 as population growth will outpace growth in the fish sector (Obiero et al., 2019). Also, the aquaculture sector in Africa remains relatively underdeveloped, resulting in limited production capacity and constrained availability of farmed aquatic products (Naylor et al., 2023). The restricted expansion of aquaculture operations contributes to the prevailing low consumption levels, as the sector fails to bridge the gap between demand and supply (Adeleke et al., 2020).

Likewise, inadequate landing facilities, substandard road networks, and poor market infrastructure pose formidable challenges (Narayan and Petesch, 2002). These deficiencies hinder the efficient movement of high-quality aquatic products across national borders within the continent, making the accessibility of diverse and nutritious aquatic foods harder and worsening the overall low consumption rates (Mindjimba et al., 2019). Furthermore, the high prevalence of postharvest losses significantly undermines the availability of safe and fresh aquatic foods in the market (Rajapaksha et al., 2021). The absence of well-developed cold chain systems intensifies this issue, resulting in the spoilage and deterioration of aquatic products during storage and transportation (Sohrabpour et al., 2012). The resultant losses in quality and quantity further limit the potential for increased consumption relative to distribution patterns.

The distribution of aquatic food varies across African countries, with coastal regions often having greater access to diverse seafood options (Figure 1). The leading five countries in terms of *per capita* daily aquatic protein supply, measured in grams, are Uganda, Egypt, Seychelles, Benin, and Mozambique, with quantities of 4.25, 3.66, 3.55, 3.55, and 3.47 grams per person per day, respectively (FAOSTAT, 2023). This highlights significant disparities in the total supply of aquatic protein among the selected countries. Uganda stands out with the highest value of 4.25 grams per person per day, indicating a comparatively higher consumption of aquatic protein sources within its population. This could be attributed to Uganda's favorable geographical location, providing abundant access to freshwater resources such as the Nile River and Lake Victoria (Yitayew and Melesse, 2011).

On the other hand, Egypt, Seychelles, Benin, and Mozambique show relatively similar levels of *per capita* aquatic protein supply, ranging from 3.47 to 3.66 grams per person per day. Despite their diverse geographical characteristics and varying proximity to marine resources, these countries display comparable consumption of aquatic protein individually. Other factors like cultural preferences, availability of alternative protein sources, and financial status may contribute to the observed similarities (Arrieta et al., 2021).

Countries that experienced relatively low food supply values (kg/ capita/yr) in 2020 include Egypt (14.98), Uganda (14.2), Malawi (9.68), Zambia (8.04), Togo (5.67), Nigeria (1.28), Djibouti (0.02), Libya (0.04), Guinea-Bissau (0.09), and São Tomé and Príncipe (0.00). These figures reveal the challenges these nations face in ensuring an adequate and sustainable supply of fish as a vital source of nutrition. Egypt, with a value of 14.98 kg/capita/yr., is one of the countries on the list with a higher food supply of freshwater fish. This can be attributed to the country's strategic location along the Nile River, which provides good opportunities for fishing activities and a consistent source of fish for its population (Abila, 2003). Uganda follows closely with a value of 14.2 kg/capita/yr. Given its large freshwater bodies, including Lake Victoria, Uganda has significant potential for fish production. However, despite the favorable conditions, the value of the food supply remains lower than expected, suggesting potential challenges in aquatic resource management (Troell et al., 2014). Malawi and Zambia, with values of 9.68 kg/capita/yr. and 8.04 kg/capita/yr., respectively, also encounter limitations in achieving optimal freshwater fish supply. Both countries have access to various lakes and rivers, such as Lake Malawi and the Zambezi River, which are known for their rich fish populations. However, factors such as poor infrastructure, inadequate fishing techniques, or socio-economic issues hinder these resources's utilization to their full potential (Mittal et al., 2010).

Countries with lower value for freshwater fish food supply include Togo, Nigeria, Djibouti, Libya, Guinea-Bissau, and São Tomé and Príncipe. Nigeria, with a value of 1.28 kg/capita/yr., is one of the most populous countries on the list. Despite its vast coastline and numerous inland waterways, Nigeria fails to fulfill its rising population's fish demand due to overfishing, insufficient processing facilities, and poor infrastructure (Ajani and Osho, 2019). Djibouti, Libya, Guinea-Bissau, São Tomé and Príncipe have considerable freshwater fish availability constraints, with values ranging from 0.02 kg/capita/yr. to 0.00 kg/ capita/yr. Due to limited water supplies, unfavorable environmental conditions, or economic restraints, some countries face challenges obtaining freshwater fish. These highlight the critical need for coordinated efforts in aquatic resource management, infrastructure development, and promoting sustainable practices. This also indicates the necessity to address the underlying issues and push for ethical and sustainable practices, greater food security, increased aquatic source food nutrition and general well-being for the people.

For marine fish food supply, Mozambique (7.18), Benin (5.73), Côte d'Ivoire (3.26), Cameroon (2.92), Gabon (2.17), and Burkina Faso (2.02) led in terms of values (kg/capita/yr). While South Sudan (0.00), Ethiopia (0.00), Chad (0.01), Zimbabwe (0.01), Guinea-Bissau (0.02), Democratic Republic of Congo (0.04), and Botswana (0.05) had relatively lower (kg/capita/yr) values (FAOSTAT, 2023). With a long fishing history and a large stretch of coastline in the Indian Ocean, Mozambique has a wealth of fisheries resources (McClanahan et al., 2011; Ollivier and Giraud, 2011). Its high consumption quantity demonstrates how much the country depends on marine fish as a source of protein. Similarly, the Benin Republic's location along the Gulf of Guinea offers the country a coastline edge, enabling the utilization of a thriving marine habitat.

Côte d'Ivoire, located on the West African coast, has a welldeveloped fishing sector and many coastal fish farmers, resulting in a comparatively high consumption of marine fish. Cameroon, another coastal country within the area, has comparable qualities and benefits from a vast range of fish species along its coastlines. Gabon, recognized for its substantial marine reserves and protected areas, consumes significant amounts of marine fish due to its natural richness (Spalding et al., 2008). Burkina Faso appears surprisingly on the list as a landlocked country. However, this value reflects a large consumption of imported marine seafood, indicating the desire for aquatic resources even in countries without direct sea access. Many countries had considerably lower numbers (kg/capita/year) for marine fish consumption. Both South Sudan and Ethiopia are landlocked countries with limited access to coastal resources, restricting their intake of marine seafood. Similarly, Chad and Zimbabwe face geographical limits that hinder seafood consumption. Despite its proximity to the sea, Guinea-Bissau faces infrastructural, governance, and economic issues that may lead to lower consumption values (Temudo and Abrantes, 2014).

The quantity of land animal protein supply surpasses aquatic protein supply across all African countries (Figure 2). The provision of protein from land animals, such as livestock, cattle, goats, sheep, and poultry, plays a significantly more prominent role in meeting the population's protein requirements when compared to aquatic sources like fish and seafood. The higher supply of land animal protein in African countries can be attributed to various factors. The longstanding integration of animal husbandry and agriculture into many African societies has contributed to domestic animal rearing and food production (McClure, 2015). Cultural and dietary preferences, as well as factors like land availability and agricultural practices, have influenced the prominence of land animals as a source of protein (Beveridge et al., 2013). Moreover, the relatively lower supply of aquatic protein may be influenced by limited access to coastal areas or freshwater bodies, lower fish production and availability, and challenges in fishery management and infrastructure development





(Reid et al., 2019). Consequently, this affects the *per capita* consumption of animal protein across the African continent.

Africa, in particular, had relatively lower per capita animal protein consumption values than other regions (Latino et al., 2020). In 2020, the United States consumed 75 g of animal protein per day, of which fish and seafood were 5.4 g. Spain consumed 68 g of animal protein, of which 12 g were fish and seafood. The United Kingdom consumed 59 g of animal protein, including 5.7 g from fish and seafood, while China consumed 39g of animal protein, including 9.4g from fish and seafood (FAOSTAT, 2023). Generally, low-income countries had low per capita availability of animal protein, averaging 13.8 grams per person per day in 2018-20. Within Africa, Southern Africa had comparatively higher values of per capita consumption of animal protein than other regions (FAOSTAT, 2023). In recent years, the evaluation of per capita intake has become more critical for understanding the nutritional profiles of different locations (Popkin, 1994). It becomes clear that, on average, people in Southern Africa prefer to consume more animal protein than their counterparts in other regions despite the diversity and variation within this region (Du Toit and Cumming, 1999).

The quantities of land animal food supply consistently exceeded aquatic food supply across all African countries (Figure 3). The continent's average quantity of land animal food significantly surpassed the average aquatic food supply. This pattern suggests a predominant reliance on land animal sources to meet dietary needs in African countries. Among the countries, Egypt stood out with the highest aquatic food supply quantity value, reaching 15.96 kg *per capita* per year. This indicates Egypt's reliance on aquatic food to fulfill its population's nutritional needs more than other African countries, underscoring the significance of fish production and distribution value chains.

Value chains play a crucial role in aquatic food consumption in tropical Africa by enhancing sustainability, economic growth through local and international trade and food security. It is vital in meeting the nutritional needs of the people, particularly in countries with a substantial reliance on aquatic sources (Belton et al., 2021). However, the volatility in aquatic food value chains has resulted in short- and long-term consequences regarding the availability and desire for aquatic foods. Hence, several components of these value chains need to be enhanced to guarantee the long-term supply and accessibility of



aquatic foods. In tropical Africa, the challenges of these value chains, intensified by the COVID-19 crisis, include risks related to economic resilience, social and environmental sustainability, changing global investment landscapes, Africa's digital transformation, and the global drive for sustainability (Feyaerts et al., 2020). Furthermore, insufficient investments in infrastructure, tropical deforestation due to economic incentives, significant rural poverty, and a lack of prioritization of tropical agriculture by governments contribute to the challenges faced by stakeholders in recent years (Belton et al., 2021). Addressing these problems requires improvements in storage, processing, and transportation infrastructure, enforcing regulations, facilitating access to finance and credit, building capacity, and providing training and technical assistance to improve production techniques and market access.

Considering the recent trends highlighted above, aquatic protein consumption is essential in the nutritional landscape, particularly in tropical Africa. To better understand this scenario, this study explored a typical relationship between aquatic protein consumption and key factors such as population, annual gross domestic product (GDP), land protein intake, aquatic food imports, and aquatic food export (Table 2). Understanding the dynamics between these variables is essential for elucidating the complex interplay of dietary habits, economic development, and trade dynamics in the context of aquatic protein consumption. The results of a multiple linear regression of total protein intake (highlighted here as a representative view) in Africa for 2020 with these factors showed that only "Import" was significant at the 10% level (p = 0.080). Increasing aquatic food imports was associated with a proportional rise in aquatic protein intake by 0.267%, suggesting a potential relationship with total protein intake. This implies explicitly that import volume may influence the consumption of aquatic protein across African countries (considering the different landlocked countries). Hence, trade dynamics may influence regional protein availability and access. Addressing this requires further studies to provide insight into the specific nature of imports and their impact on aquatic protein consumption in the region. Also, further insights from trade economics of aquatic foods

Total protein intake	Coefficient	Standard error	t	P >  t	[95% Conf. Interval]	
GDP	0.0004	0.0019	0.23	0.819	-0.0034	0.0043
Import	0.0027	0.0015	1.80	0.080	-0.0003	0.0057
Export	-0.0007	0.0009	-0.81	0.422	-0.0025	0.0011
Land protein intake	0.02789	0.0306	0.91	0.367	-0.0339	0.0898
Population	-6.40e-06	6.68e-06	-0.96	0.344	-0.00002	7.10e-06

TABLE 2 Multiple linear regression for aquatic protein intake in Africa with some key factors for 2020.

across the continent would help stakeholders and policymakers to develop targeted strategies to enhance aquatic protein access and utilization in Africa.

Other variables such as land protein intake, export, GDP, and population showed a non-significant relationship. This indicates that these factors may have limited direct influence on total protein, at least for the year under study. Given the non-significant relationships observed in the regression analysis, there is a need for more comprehensive research to identify additional factors that may influence aquatic protein consumption in Africa. Specifically, it is essential to consider the time course dynamics of these factors and their influence on aquatic protein consumption, as other non-significant factors for 2020 may be significant in the future. Thus, a holistic and time-course approach is required to draw valid conclusions.

Generally, understanding seafood demand patterns is of significant importance in research. However, existing literature has primarily concentrated on developed nations and commercially important fish species, resulting in a dearth of information regarding consumer preferences for fish in many African countries (Obiero et al., 2019). While previous investigations have considered consumer preferences and future demand changes, there are limited details, particularly concerning the African context [Food and Agriculture Organization (FAO), 2017a,b]. Of particular note is the dearth of studies that have effectively integrated the complex relationship between urbanization, income growth, and accurate future fish demand estimations (Béné et al., 2015). Thus, studies have relied on assumptions of constant consumption rates in the future (Barange et al., 2014) or fixed nutritional targets (Rice and Garcia, 2011) without considering the dynamic nature of dietary needs and evolving nutritional requirements over time. Addressing these multifaceted challenges requires comprehensive and strategic interventions (Adelman and Taylor, 2002). Enhancing capture fisheries management practices, such as implementing sustainable fishing methods and ensuring responsible resource utilization, can help mitigate the supply-demand imbalance (Gutierrez, 2015). Simultaneously, fostering the growth of the aquaculture sector through investments in research, technology, and infrastructure is imperative to bolster production and expand availability (Nguyen, 2016; Elegbeleye et al., 2024). Furthermore, addressing the deficiencies in landing, road, and market infrastructures necessitates substantial investments in improving logistics and transportation networks [Food and Agriculture Organization (FAO), 2022a].

This strategy would enable the seamless movement of aquatic products across borders, facilitating greater access to diverse markets and enhancing consumption opportunities. Also, prioritizing the development of robust cold chain systems is vital for minimizing post-harvest losses. Investments in storage facilities, refrigeration technologies, and efficient distribution networks would ensure the preservation of product quality and enhance consumer confidence, leading to increased consumption (Sugri et al., 2021). In relation to this, it is imperative to develop all-encompassing policy frameworks to foster sustainable and nourishing consumption of aquatic food in Africa. Such policy frameworks entail integrating fisheries and aquaculture management strategies with broader food security, nutrition, and socio-economic development initiatives (Béné et al., 2016). By fostering an enabling environment that encourages innovation, research, and collaboration among stakeholders, Africa can unlock the full potential of its aquatic resources and achieve improved consumption patterns in the future (Tigchelaar et al., 2022).

# 4 Barriers/challenges to adequate consumption of aquatic foods in tropical Africa

Aquaculture, a burgeoning industry in tropical Africa, holds tremendous promise as an alternative protein source and a complement to traditional fisheries [Food and Agriculture Organization (FAO), 2018]. However, significant obstacles to expansion exist within the industry, including restricted access to finance, poor infrastructure, and a lack of technical experience (George et al., 2016). Despite the potential benefits of aquatic food production, many local communities encounter significant barriers to accessing and consuming aquatic foods. These challenges take several forms due to economic, social, and cultural variables (Baumgartner et al., 2021).

Economically, limited access to financial resources and market possibilities has hindered aquatic food production and distribution, particularly for small-scale farmers (Béné et al., 2019). Socially and culturally, a preference for land-based animal products or a lack of awareness regarding the nutritional benefits of aquatic foods has contributed to low consumption levels (Cochrane et al., 2016). Furthermore, environmental factors such as water pollution, habitat destruction, and climate change have adversely affected the availability and quality of aquatic food sources (Khan et al., 2021). Poverty further limits the sufficient consumption of aquatic foods, and many communities in the region have been economically marginalized, with limited access to resources like income, education, and healthcare.

Consequently, they struggle to afford nutritious foods, including aquatic foods, which are costlier than other protein sources (FAO and ECA, 2018). Besides, insufficient infrastructure, characterized by inadequate transport networks, storage facilities, and processing facilities, has significantly constrained the availability and quality of

aquatic foods across various regions (Rolle, 2006). Therefore, communities encounter difficulties in accessing and consuming these foods. Inadequate water and sanitation systems further worsen the issue by compromising the safety and quality of aquatic foods. Such foods sometimes become contaminated with bacteria and other harmful substances, thus undermining their suitability for consumption (Jennings et al., 2016). These social obstacles can be linked to many factors arising from complex interactions among socio-economic elements, gender gaps, cultural norms, and restricted availability of resources. Socio-economic disparities, gender inequalities, limited access to quality water sources, and a lack of awareness regarding the nutritional value of fish are examples of these challenges. For instance, lack of access to improved water services, reliance on poor energy sources like biomass, and access to quality water impact the population's health and contribute to food insecurity (Nwozor et al., 2019; Fayad, 2023).

Cultural factors also limit the consumption of aquatic foods in tropical Africa. For example, a lack of awareness regarding the nutritional significance of fish, particularly within specific societal groups, can hinder the incorporation of aquatic foods into regular diets. Also, traditional beliefs or practices may influence dietary choices and limit fish consumption and other aquatic foods. Notably, specific communities prefer terrestrial protein sources, such as beef or chicken, due to deeply ingrained cultural or religious beliefs (Looy et al., 2014). Additionally, social norms governing food preparation and consumption may discourage the utilization of aquatic foods, particularly among women who commonly assume the responsibility for food preparation in many communities (Meyer-Rochow, 2009).

Climate change is another significant barrier to adequate consumption of aquatic foods in tropical Africa. The escalating sea temperatures and shifting weather patterns have deleterious effects on the productivity and distribution of fish stocks, thus impeding communities' access to and consumption of these vital food sources (Barange et al., 2018). Consequently, effectively addressing the complex challenges associated with aquatic food consumption necessitates concerted efforts and comprehensive solutions that acknowledge the multifaceted nature of the barriers hindering access to and consumption of aquatic foods (Elkin and Katz, 2019). Thus, harnessing the vast potential of aquatic food production in tropical Africa can benefit the region's people, economies, and environment.

A notable endeavor in the field involves implementing various initiatives to foster responsible aquaculture practices. For instance, the Aquaculture Stewardship Council (ASC) has emerged as a significant organization offering certification to ensure adherence to responsible aquaculture methods (Kruk and Peters, 2018). The ASC certification is pivotal in promoting responsible aquaculture practices, ensuring sustainability and environmental stewardship within the industry. However, the low ASC certification rates, particularly in the Global South and Africa, present significant challenges that must be addressed to foster widespread adoption of responsible aquaculture methods (Bush et al., 2013). This underscores the importance of understanding and addressing the barriers hindering certification uptake in these underrepresented areas. One key barrier identified is the lack of awareness and understanding of ASC certification standards and benefits among African aquaculture stakeholders. Limited access to information and insufficient outreach efforts contribute to this challenge, inhibiting the widespread adoption of certification practices. Moreover, high certification costs are challenging for aquaculture producers in these regions, where financial resources may be limited. This calls for targeted interventions, including financial support mechanisms, capacity-building programs, infrastructure development projects, and policy initiatives incentivizing certification to address these challenges and improve African ASC certification rates (Bush et al., 2013).

Such interventions could be spearheaded by organizations such as the Aquaculture Network for Africa (ANAF), which has been established to facilitate technical assistance and promote the exchange of information among stakeholders within the sector (Nath et al., 2000). Also, the African Union's Policy Framework and Reform Strategy for Fisheries and Aquaculture in Africa, along with the Sustainable Fisheries Management Project, supported by the World Bank, contribute substantially to the comprehensive development of the sector (AUC-NEPAD, 2014). These initiatives emphasize the enhancement of governance structures, the implementation of responsible fishing techniques, and the augmentation of investments in the sector (Franz et al., 2019). Collectively, these efforts can enhance fishery management in the region by bolstering legal and institutional frameworks, promoting sustainable fishing practices, and improving the livelihoods of individuals engaged in the fishing industry.

# 5 Sustainability of aquatic food production in Africa and implications on health

Sustainable aquatic food production is critical in ensuring food security, fostering economic development, and promoting environmental conservation in Africa (El Bilali et al., 2019). The African continent boasts an extensive water resource base comprising rivers, lakes, and coastal areas, which present abundant opportunities to produce diverse aquatic foods, including fish, crustaceans, and molluscs (Zilhão et al., 2020). However, several factors pose significant threats to the sustainability of aquatic food production in Africa, including overfishing, weak governance, climate change, and pollution (Ahern et al., 2021).

Overfishing is one of the most pressing concerns among these challenges, jeopardizing the sustainability of aquatic food production in Africa (Lucrezi, 2022). Unsustainable fishing practices have severely depleted numerous regional fish stocks (Haque et al., 2021). This depletion threatens the viability of fish stocks and the livelihoods of millions of individuals who rely on them for sustenance and income (Bi et al., 2022). In tropical Africa, overfishing, characterized by unsustainable fishing techniques like using destructive gear and targeting immature fish, has depleted various fish stocks (Srinivasan et al., 2012). The depletion of fish stocks due to overfishing represents a critical challenge to the sustainability of aquatic food production in Africa, with far-reaching consequences for local communities (Ikram et al., 2023). This situation directly impacts the availability and affordability of fish, limiting their access to vital sources of nutrition (Ansari et al., 2021). Fish is a good source of omega-3 fatty acids essential for cardiovascular health, brain growth, and cognitive function (Sanjay Panda and Diwan, 2023). A decrease in fish supply may raise the risk of malnutrition, cardiovascular diseases, and other conditions linked to omega-3 deficiency. Several countries in tropical Africa lack the key institutional and legal frameworks for aquatic

resources management (DiNicolantonio and O'Keefe, 2020). Specifically, poor governance poses a significant barrier to successfully managing fisheries and the sustainability of aquaculture food production in Africa (Wilson and McCay, 2019). Many countries within the region face the absence of necessary legal and institutional structures, resulting in inadequate enforcement of regulations, the prevalence of illegal fishing practices, and conflicts among various user groups (Begum et al., 2022). This dearth of governance also extends to aquaculture management, wherein deficient monitoring and regulation exacerbate pollution and the spread of diseases (Edokpayi et al., 2017).

Moreover, the impact of climate change intensifies the issues mentioned above. The increasing sea temperatures, ocean acidification, and shifting weather patterns affect the productivity and distribution of fish stocks (Maulu et al., 2021; Lubembe et al., 2022). Extreme weather events, including floods and droughts, cause infrastructure damage and disrupt production systems. Also, the rising sea levels threaten the viability of coastal aquaculture (Gitz et al., 2016), consequently impacting the availability and affordability of fish resources (Katikiro and Macusi, 2012). Water temperature and salinity changes also affect the quality and safety of aquatic foods, as warmer waters facilitate the growth of harmful bacteria and toxins (Gomez-Zavaglia et al., 2020). Furthermore, climate change-induced extreme weather events, such as floods and droughts, directly affect the accessibility and safety of aquatic foods, thereby contributing to malnutrition and various health issues (Duchenne-Moutien and Neetoo, 2021).

Besides climate change, aquatic pollution significantly threatens the sustainability of aquatic food production in Africa (Mpumi et al., 2020). The contamination of water bodies resulting from industrial and agricultural activities, as well as untreated sewage and waste disposal, has detrimental effects on the safety and nutritional quality of aquatic foods (Mateo-Sagasta et al., 2017). Consequently, aquatic pollution has led to the accumulation of harmful substances, including antibiotics and heavy metals, in fish and other aquatic organisms, posing potential risks to human health (Okocha et al., 2018; Qadri and Faiq, 2020; Okon et al., 2022). In this regard, consuming fish contaminated with mercury can result in neurological damage and developmental disorders. Similarly, exposure to polycyclic aromatic hydrocarbons (PAHs) may elevate the risk of cancer (Mitra et al., 2022) and require a more proactive approach to tackle aquatic organisms and human exposure.

Various initiatives have been implemented to address these challenges and promote the sustainability of aquatic food production in Africa. These efforts encompass improvements in governance, exemplified by the African Union's Policy Framework and Reform Strategy for Fisheries and Aquaculture in Africa. This framework provides a comprehensive structure for the sector's sustainable development, focusing on enhanced governance, responsible fishing practices, and increased investments (NEPAD Planning and Coordinating Agency, 2016). Other initiatives concentrate on advancing sustainable fishing practices. For example, the Sustainable Fisheries Management Project, backed by the World Bank, strengthens legal and institutional frameworks, encourages sustainable fishing practices, and enhances fishers' livelihoods (Guenard, 2021). These initiatives generally work toward a shared objective of maximizing the potential of aquatic resources in the region.

Africa has vast untapped potential for advancing aquatic food production (De Angelis et al., 2021). The continent's copious freshwater resources and favorable climatic conditions establish an ideal setting for cultivating different aquatic species. For instance, aquaculture can generate income in rural communities, alleviate strain on wild fish populations, and enhance food production (Odende et al., 2022). By prioritizing investments in research, development, infrastructure, and capacity-building initiatives, African countries can effectively harness the potential of aquatic food production to meet their increasing demands, concurrently addressing poverty reduction and stimulating economic growth (Ali et al., 2021). However, realizing the full potential of aquatic food resources in Africa necessitates unified efforts and collaboration among governments, stakeholders, and international organizations (Zhao et al., 2022). It, therefore, becomes imperative to actively promote more sustainable practices, enhance infrastructure, facilitate market access, and fortify institutional frameworks to bolster the expansion of these sectors.

The Aquaculture Network for Africa (ANAF), one of such institutional frameworks, is essential in advancing sustainable aquaculture in Africa. The ANAF functions as a collaborative platform that facilitates technical assistance and information exchange among stakeholders within the aquaculture industry (Capobianco et al., 2021). Through the promotion of knowledge sharing, the ANAF empowers farmers, researchers, policymakers, and other crucial participants to gain access to valuable insights, adopt best practices, and explore innovative solutions (Vermunt et al., 2022). This networkbased approach fosters the adoption of sustainable aquaculture methods, expedites technological progress, and facilitates sharing of resources and expertise across different African regions (Reddy et al., 2021). Apart from the Aquaculture Stewardship Council (ASC) and the Asian Aquaculture Network (AAN), numerous other organizations, governmental bodies, and research institutions contribute to developing and expanding sustainable aquaculture in Africa. These efforts encompass various activities, including promoting capacity-building, encouraging infrastructure investment, researching local species and ecosystems, and facilitating market access for aquatic foods. By addressing these diverse aspects, the collective initiatives aim to establish a flourishing and sustainable aquaculture sector that effectively complements traditional fisheries and provides a sustainable source of protein for communities throughout the region (Brears, 2021).

The sustainability of aquatic food production in Africa plays a significant role in determining the price, consumption patterns, and dietary intake of aquatic foods, thereby impacting human health (Golden et al., 2021). The availability and affordability of fish and other aquatic foods are threatened by unsustainable fishing methods, pollution, and climate change, which leads to an increase in price (Agostoni et al., 2023). This increase in price makes it challenging for low-income communities to access nutritious aquatic foods, leading to reduced consumption and a shift toward more affordable but less nutritious food options (Belton et al., 2020). The accessibility and cost of fish and other aquatic foods directly impact consumer intake and human health outcomes (Fiorella et al., 2021). Fish availability declines due to overfishing or pollution, decreasing fish consumption and resulting in insufficient vital nutrients like omega-3 fatty acids (Gormaz et al., 2014).

Moreover, the price of aquatic foods might discourage intake, especially for low-income groups that frequently have trouble accessing other proteins and minerals (Maulu et al., 2021). Access to and availability of aquatic foods is also critical for the health and wellbeing of millions of people in Africa. Promoting and emphasizing sustainable practices could ensure fish supplies' long-term safety and nutritional value (Khan et al., 2021). This can be accomplished by enacting rules and regulations that promote sustainable fishing practices, such as quotas and gear limits.

Furthermore, efforts are required to reduce the adverse effects of climate change on aquatic ecosystems and regulate pollution (Odeku and Paulos, 2017). Aquatic ecosystems can be challenged by overfishing, pollution, and climate change while still gaining access to the advantages of aquatic foods (Zhang et al., 2023). Implementing appropriate management measures will eventually improve the health and well-being of millions of Africans (Andrews et al., 2021). These actions will require raising more awareness among local communities about the importance of the safe and healthy consumption of aquatic foods. Furthermore, supporting research on these foods' nutritional quality and safety can guarantee meeting the necessary standards (Khan et al., 2021), while creating more opportunities.

Aquatic food production opportunities and sustainability differ among coastal and landlocked African countries. Coastal countries have the advantage of access to the ocean, allowing them to utilize various marine resources such as fish and shellfish (Bennett et al., 2020). Coastal African countries typically have access to the marine biodiversity that offers valuable fish species. The ocean environment provides high protein content, important fatty acids and essential micronutrients, making seafood an integral part of their diets (Aakre et al., 2020). On the other hand, landlocked countries rely on rivers, lakes and inland fisheries for their food needs (Kolding et al., 2019).

Moreover, coastal regions are usually better positioned for seafood trade as they can export and generate income through this industry. At the same time, landlocked African countries face the challenge of being geographically distant from the coast (Ferguson, 2021). These inland resources can be equally rich; however, their sustainability depends on water quality management, effective fisheries management practices, transportation infrastructure availability, and storage purposes (Anselem Tengecha and Zhang, 2020). Coastal countries naturally have an advantage regarding fisheries-based industries and tourism related to resources. They can quickly develop their aquaculture production and processing sector for export purposes (Uddin et al., 2021). The preferences and dietary traditions also vary as coastal communities tend to have an affinity for seafood, while landlocked communities may have practices centered around other food sources (Cooke et al., 2018; Ferguson, 2021).

For landlocked countries to thrive, they must diversify their economies by engaging in activities such as aquaculture, utilizing irrigation techniques and adopting agricultural practices incorporating aquatic resources (Das and Mandal, 2022). Thus, a collaboration between landlocked countries can be mutually beneficial when harnessing the full potential of aquatic food resources. In essence, coastal countries can focus on implementing fishing practices, while landlocked countries can explore partnerships and agreements to access shared water bodies. Taking these aspects into account is crucial when formulating policies and managing resources.

# 6 Recent aquatic policies to make aquatic foods more accessible

Policy interventions are crucial in promoting sustainable aquatic food systems by addressing environmental, social, and

economic challenges (Ahern et al., 2021). The International Food Policy Research Institute (IFPRI) discussed policy issues and options in the Aquatic Food System (Ragasa and Loison, 2023). Some policy interventions in aquatic food systems highlighted include implementing community-based fisheries management for increased fish stocks and improved livelihoods for fishing communities. Also, establishing a national aquaculture development plan and introducing a fishery co-management system, directed by comprehensive guidelines can increase investment and growth in the aquaculture sector and improve fishery resource conservation and sustainable use, respectively (Kyvelou et al., 2023).

Given this, the Food and Agriculture Organization (FAO) (2021) has developed guidelines for sustainable aquaculture, which provide a framework for responsible and sustainable aquaculture practices. One such intervention prioritizes aquatic foods for domestic consumption over export, particularly in areas with high malnutrition rates (Farmery et al., 2020). This can be achieved through policies that support local production and distribution of aquatic foods and measures to reduce trade barriers and increase market access for small-scale producers (Wood et al., 2021). Another intervention is to improve the food safety of aquatic foods through improved governance at all levels, as well as behavioral and systemic changes, such as enabling a better circular economy framework and more sustainable production and consumption patterns (Cooney et al., 2023). Prioritizing the revision of existing regulatory frameworks, institutional arrangements, and other instruments related to marine litter and their enforcement can also help identify synergies, gaps and potential solutions globally and regionally, thus reducing and avoiding impacts on aquatic food systems and consumers (Graham, 2023). Finally, policies that support research and development of new technologies for sustainable aquaculture practices can also help increase the availability of aquatic foods (Gephart et al., 2021).

## 7 Limitations of the study

The broad scope of this study has both strengths and limitations. On the one hand, it allows for an overview of aquatic food resources in tropical Africa, which serves as a basis for more localized studies in the future. However, this broad nature is limited when making recommendations and policy implications for countries or regions. Therefore, exploring research directions focusing on localized and context-specific investigations to address this limitation is vital. This will offer tailor-made solutions, focused policy recommendations, and strategies directly impacting the regions or communities involved.

Besides, tropical African countries have diverse aquatic ecosystems and communities, each with distinct cultures, characteristics and challenges. Given the nature of this study, there is a possibility that it may not fully capture these differences. Therefore, future research should consider the sociocultural variations within these regions, as these factors can influence the applicability of recommendations and policies. Another approach is to promote collaborative research endeavors by combining the expertise of ecologists, economists, anthropologists and policymakers to devise solutions that tackle the dynamics of aquatic food resources in tropical Africa based on specific contexts.

## 8 Conclusion and future directions

In conclusion, this research offers significant insights into the quantities of aquatic protein supplied across tropical African countries. It provides valuable insights into these countries' aquatic source food consumption patterns. The results underscore the differences in aquatic protein consumption among the countries, fostering the need to consider local factors when developing nutrition interventions and policies. This study showed the dynamics of local culture, socio-economic status, environmental conditions, and individual dietary behaviors, offering actionable recommendations for tailoring interventions to optimize effectiveness within diverse African communities. In this regard, future research is recommended to investigate the specific origins of aquatic protein and its role in the overall dietary landscape of these nations, thereby facilitating targeted approaches to enhance food security and public health outcomes.

This research further demonstrates the pattern of an aquatic protein supply, which showed lower values than land animal protein supply across African countries. These findings highlight the significance of land-based animal sources in meeting the population's protein needs. Specifically, aquatic food imports significantly influenced their consumption across the region for the year under study, suggesting the influence of trade economics on aquatic food intake. In this regard, a clearer understanding of these factors influencing the differences and their dynamics is essential for policymakers, nutritionists, and stakeholders in devising targeted strategies to enhance protein availability and address imbalances between land and aquatic protein sources in African countries. Therefore, further research is required to investigate the underlying factors contributing to these gaps and explore potential solutions for sustainable aquatic source food production and consumption in the region.

Furthermore, the production of aquatic food in tropical Africa is an intricate and diversified sector of the economy that supports

### References

Aakre, I., Bøkevoll, A., Chaira, J., Bouthir, F. Z., Frantzen, S., Kausland, A., et al. (2020). Variation in nutrient composition of seafood from North West Africa: Implications for food and nutrition security. *Foods*, 9:1516.

Abegaz, K. H. (2018). Prevalence of undernourishment: trend and contribution of east African countries to sub-Saharan Africa from 1991 to 2015. *Agric. Food Secur.* 7, 1–6. doi: 10.1186/s40066-018-0198-9

Abila, R. O. (2003). Fish trade and food security: are they reconcilable in Lake Victoria. *Kenya Marine Fisheries* 31:49.

Adeleke, B., Robertson-Andersson, D., Moodley, G., and Taylor, S. (2020). Aquaculture in Africa: a comparative review of Egypt, Nigeria, and Uganda Vis-a-Vis South Africa. *Rev. Fisheries Sci. Aquacult.* 29, 167–197. doi: 10.1080/23308249.2020.1795615

Adelman, H. S., and Taylor, L. (2002). Building comprehensive, multifaceted, and integrated approaches to address barriers to student learning. *Child. Educ.* 78, 261–268. doi: 10.1080/00094056.2002.10522738

Agostoni, C., Baglioni, M., La Vecchia, A., Molari, G., and Berti, C. (2023). Interlinkages between climate change and food systems: the impact on child malnutrition—narrative review. *Nutrients* 15:416. doi: 10.3390/nu15020416

Ahern, M., Thilsted, S., Oenema, S., and Kühnhold, H. (2021). The role of aquatic foods in sustainable healthy diets. *UN Nutrition Discussion Paper*.

Ajani, E. K., and Osho, F. E. (2019). Diagnosis and management of fishery problems, flooding and value addition in fish production in Nigeria. In Proceedings of the southwest zonal Refils sectorial meeting on OFAR and extension held at the Institute of Agricultural Research and Training, moor-plantation, Ibadan, Nigeria.

Akegbejo-Samsons, Y. (2022). "Aquaculture and fisheries production in Africa: highlighting potentials and benefits for food security" in *Food security for African smallholder farmers*. eds. H. A. Mupamba, A. D. Nciizah, P. Nyambo, B. Muchara, and N. N. Gabriel (Springer Nature Singapore: Singapore), 171–190.

millions of people. The sector involves a variety of fishing, aquaculture, and fish farming techniques. However, the sector has experienced challenges and barriers, including poor infrastructure, inadequate legislation, and restricted access to finance. Diverse approaches have been proposed to address these issues, such as increased financial accessibility and environmentally friendly aquaculture practices. However, implementing these approaches requires more practical efforts and ongoing monitoring to ensure effective outcomes for a healthier African population.

### Author contributions

EO: conceptualization, formal analysis, and writing – original draft preparation. RO: formal analysis and writing – reviewing and editing. DF, PO, and BF: writing – original draft preparation. All authors contributed to the article and approved the submitted version.

### **Conflict of interest**

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

### Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Aldaya, M. M., Ibañez, F. C., Domínguez-Lacueva, P., Murillo-Arbizu, M. T., Rubio-Varas, M., Soret, B., et al. (2021). Indicators and recommendations for assessing sustainable healthy diets. *Food Secur.* 10:999. doi: 10.3390/foods10050999

Ali, E. B., Anufriev, V. P., and Amfo, B. (2021). Green economy implementation in Ghana as a road map for a sustainable development drive: a review. *Scientific African* 12:e00756. doi: 10.1016/j.sciaf.2021.e00756

Andrews, N., Bennett, N. J., Le Billon, P., Green, S. J., Cisneros-Montemayor, A. M., Amongin, S., et al. (2021). Oil, fisheries and coastal communities: a review of impacts on the environment, livelihoods, space and governance. *Energy Res. Soc. Sci.* 75:102009. doi: 10.1016/j.erss.2021.102009

Ansari, F. A., Guldhe, A., Gupta, S. K., Rawat, I., and Bux, F. (2021). Improving the feasibility of aquaculture feed by using microalgae. *Environ. Sci. Pollut. Res.* 28, 43234–43257. doi: 10.1007/s11356-021-14989-x

Anselem Tengecha, N., and Zhang, X. (2020). Status, Constraints and Strategies of Marine Traffic Flow on Dar es Salaam Port, Tanzania. In *Proceedings of the 2020 2nd International Conference on Robotics Systems and Vehicle Technology*. (pp. 54–59).

Arrieta, E. M., Geri, M., Coquet, J. B., Scavuzzo, C. M., Zapata, M. E., and González, A. D. (2021). Quality and environmental footprints of diets by socioeconomic status in Argentina. *Sci. Total Environ.* 801:149686. doi: 10.1016/j. scitotenv.2021.149686

Arshad, N., Samat, N., and Lee, L. K. (2022). Insight into the relation between nutritional benefits of aquaculture products and its consumption hazards: a global viewpoint. *Front. Mar. Sci.* 9:925463. doi: 10.3389/fmars.2022.925463

AUC-NEPAD (2014). The policy framework and reform strategy for fisheries and aquaculture in Africa. African Union Commission and NEPAD Planning and Coordinating Agency.

Awuchi, C. G., Ondari, E. N., Nwozo, S., Odongo, G. A., Eseoghene, I. J., Twinomuhwezi, H., et al. (2022). Mycotoxins' toxicological mechanisms involving humans, livestock and their associated health concerns: a review. *Toxins* 14:167. doi: 10.3390/toxins14030167

Barange, M., Bahri, T., Beveridge, M. C., Cochrane, K. L., Funge-Smith, S., and Poulain, F. (2018). Impacts of climate change on fisheries and aquaculture. *United Nations' Food Agric. Organ.* 12, 628–635.

Barange, M., Merino, G., Blanchard, J. L., Scholtens, J., Harle, J., Allison, E. H., et al. (2014). Impacts of climate change on marine ecosystem production in societies dependent on fisheries. *Nat. Clim. Chang.* 4, 211–216. doi: 10.1038/nclimate2119

Baumgartner, L. J., Barlow, C., Mallen-Cooper, M., Boys, C., Marsden, T., Thorncraft, G., et al. (2021). Achieving fish passage outcomes at irrigation infrastructure; a case study from the lower Mekong Basin. *Aquac. Fisheries* 6, 113–124. doi: 10.1016/j. aaf.2018.12.008

Begum, A., Uddin, M. K., Rahman, M. M., Shamsuzzaman, M. M., and Islam, M. M. (2022). Assessing policy, legal and institutional frameworks of marine megafauna conservation in Bangladesh. *Mar. Policy* 143:105187. doi: 10.1016/j.marpol.2022.105187

Belton, B., Little, D. C., Zhang, W., Edwards, P., Skladany, M., and Thilsted, S. H. (2020). Farming fish in the sea will not nourish the world. *Nat. Commun.* 11:5804. doi: 10.1038/s41467-020-19679-9

Belton, B., Rosen, L., Middleton, L., Ghazali, S., Mamun, A. A., Shieh, J., et al. (2021). COVID-19 impacts and adaptations in Asia and Africa's aquatic food value chains. *Mar. Policy* 129:104523. doi: 10.1016/j.marpol.2021.104523

Béné, C., Arthur, R., Norbury, H., Allison, E. H., Beveridge, M., Bush, S., et al. (2016). Contribution of fisheries and aquaculture to food security and poverty reduction: assessing the current evidence. *World Dev.* 79, 177–196. doi: 10.1016/j. worlddev.2015.11.007

Béné, C., Barange, M., Subasinghe, R., Pinstrup-Andersen, P., Merino, G., Hemre, G. I., et al. (2015). Feeding 9 billion by 2050–putting fish back on the menu. *Food Secur.* 7, 261–274. doi: 10.1007/s12571-015-0427-z

Béné, C., Oosterveer, P., Lamotte, L., Brouwer, I. D., de Haan, S., Prager, S. D., et al. (2019). When food systems meet sustainability-current narratives and implications for actions. *World Dev.* 113, 116–130. doi: 10.1016/j.worlddev.2018.08.011

Bennett, N. J., Finkbeiner, E. M., Ban, N. C., Belhabib, D., Jupiter, S. D., Kittinger, J. N., et al. (2020). The COVID-19 pandemic, small-scale fisheries and coastal fishing communities. *Coastal management*, 48, 336–347.

Beveridge, M. C., Thilsted, S. H., Phillips, M. J., Metian, M., Troell, M., and Hall, S. J. (2013). Meeting the food and nutrition needs of the poor: the role of fish and the opportunities and challenges emerging from the rise of aquaculturea. *J. Fish Biol.* 83, 1067–1084. doi: 10.1111/jfb.12187

Bi, M., Zhang, Z., Guo, X., and Wan, L. (2022). Evaluation of sustainable utilization of African marine fishery resources. *Aust. Fish.* 8:4. doi: 10.3390/fishes8010004

Bianchi, M., Hallström, E., Parker, R. W., Mifflin, K., Tyedmers, P., and Ziegler, F. (2022). Assessing seafood nutritional diversity together with climate impacts informs more comprehensive dietary advice. *Commun. Earth Environ.* 3:188. doi: 10.1038/ s43247-022-00516-4

Boyd, C. E., McNevin, A. A., and Davis, R. P. (2022). The contribution of fisheries and aquaculture to the global protein supply. *Food Secur.* 14, 805–827. doi: 10.1007/s12571-021-01246-9

Bradley, B., Byrd, K. A., Atkins, M., Isa, S. I., Akintola, S. L., Fakoya, K. A., et al. (2020). Fish in food systems in Nigeria: A review. Monographs.

Brears, R. C. (2021). Developing the blue economy. London: Palgrave Macmillan.

Bush, S. R., Belton, B., Hall, D., Vandergeest, P., Murray, F. J., Ponte, S., et al. (2013). Certify sustainable aquaculture? *Science* 341, 1067–1068. doi: 10.1126/science.1237314

Calder, P. C. (2013). "Nutritional benefits of omega-3 fatty acids" in *Food enrichment with omega-3 fatty acids*. eds. C. Jacobsen, N. Skall Nielsen, A. Frisenfeldt Horn, and A-D. Moltke Sorensen (Cambridge: Woodhead Publishing), 3–26.

Calder, P. C. (2021). "Health benefits of omega-3 fatty acids" in *Omega-3 Delivery Systems*. eds. P. J. Garcia-Moreno, C. Jacobsen, A-D. Moltke Soresen, and B. Yesiltas (Cambridge, MA: Academic Press), 25–53.

Capobianco, N., Basile, V., Loia, F., and Vona, R. (2021). Toward a sustainable decommissioning of offshore platforms in the oil and gas industry: a PESTLE analysis. *Sustain. For.* 13:6266. doi: 10.3390/su13116266

Chan, C. Y., Tran, N., Cheong, K. C., Sulser, T. B., Cohen, P. J., Wiebe, K., et al. (2021). The future of fish in Africa: employment and investment opportunities. *PLoS One* 16:e0261615. doi: 10.1371/journal.pone.0261615

Chan, C. Y., Tran, N., Pethiyagoda, S., Crissman, C. C., Sulser, T. B., and Phillips, M. J. (2019). Prospects and challenges of fish for food security in Africa. *Glob. Food Sec.* 20, 17–25. doi: 10.1016/j.gfs.2018.12.002

Cochrane, S. K., Andersen, J. H., Berg, T., Blanchet, H., Borja, A., Carstensen, J., et al. (2016). What is marine biodiversity? Towards common concepts and their implications for assessing biodiversity status. *Front. Mar. Sci.* 3:248. doi: 10.3389/fmars.2016.00248

Cooke, S. J., Twardek, W. M., Lennox, R. J., Zolderdo, A. J., Bower, S. D., Gutowsky, L. F., et al. (2018). The nexus of fun and nutrition: Recreational fishing is also about food. *Fish and fisheries*, 19, 201–224.

Cooney, R., de Sousa, D. B., Fernández-Ríos, A., Mellett, S., Rowan, N., Morse, A. P., et al. (2023). A circular economy framework for seafood waste valorization to meet challenges and opportunities for intensive sustainability. *J. Clean. Prod.* 392:136283. doi: 10.1016/j.jclepro.2023.136283

Cowx, I. G., and Ogutu-Owhayo, R. (2019). Towards sustainable fisheries and aquaculture management in the African Great Lakes. *Fish. Manag. Ecol.* 26, 397–405. doi: 10.1111/fme.12391

Craig, W. J., Mangels, A. R., Fresán, U., Marsh, K., Miles, F. L., Saunders, A. V., et al. (2021). The safe and effective use of plant-based diets with guidelines for health professionals. *Nutrients* 13:4144. doi: 10.3390/nu13114144

Das, S. K., and Mandal, A. (2022). Diversification in aquaculture resources and practices for smallholder farmers. In *Agriculture, Livestock Production and Aquaculture: Advances for Smallholder Farming Systems* Volume 1 (pp. 263–286). Cham: Springer International Publishing.

De Angelis, P., Tuninetti, M., Bergamasco, L., Calianno, L., Asinari, P., Laio, F., et al. (2021). Data-driven appraisal of renewable energy potentials for sustainable freshwater production in Africa. *Renew. Sust. Energ. Rev.* 149:111414. doi: 10.1016/j. rser.2021.111414

Deme, E., Deme, M., Failler, P., Bocoum, W., Diédhiou, I., Touron-Gardic, G., et al. (2023). Contribution of small-scale migrant fishing to the emergence of the fishmeal industry in West Africa: cases of Mauritania, Senegal and the Gambia. *Front. Mar. Sci.* 10:871911. doi: 10.3389/fmars.2023.871911

DiNicolantonio, J. J., and O'Keefe, J. H. (2020). The importance of marine omega-3s for brain development and the prevention and treatment of behavior, mood, and other brain disorders. *Nutrients* 12:2333. doi: 10.3390/nu12082333

Du Toit, J. T., and Cumming, D. H. (1999). Functional significance of ungulate diversity in African savannas and the ecological implications of the spread of pastoralism. *Biodivers. Conserv.* 8, 1643–1661. doi: 10.1023/A:1008959721342

Duchenne-Moutien, R. A., and Neetoo, H. (2021). Climate change and emerging food safety issues: a review. J. Food Prot. 84, 1884–1897. doi: 10.4315/JFP-21-141

Edokpayi, J. N., Odiyo, J. O., and Durowoju, O. S. (2017). Impact of wastewater on surface water quality in developing countries: a case study of South Africa. *Water quality* 10, 10–5772. doi: 10.5772/66561

El Bilali, H., Callenius, C., Strassner, C., and Probst, L. (2019). Food and nutrition security and sustainability transitions in food systems. *Food Energy Security* 8:e00154. doi: 10.1002/fes3.154

Elegbeleye, J., Ntuli, V., Kofi Agbemavor, W., and Adeyanju, A. (2024). "Chapter 23 enhancement of food safety through technological innovations" in *Food safety and toxicology: Present and future perspectives*. eds. O. Ijabadeniyi and O. Folake Olagunju (Berlin: De Gruyter), 455–502.

Elkin, Z., and Katz, I. (2019). Implementation of the sustainable development goals: national review. *Global Risk Insights* 430, 1–429.

Eyayu, A., Getahun, A., and Keyombe, J. L. (2023). A review of the production status, constraints, and opportunities in east African freshwater capture and culture fisheries. *Aquac. Int.* 31, 2057–2078. doi: 10.1007/s10499-023-01071-1

FAO and ECA (2018). Regional Africa overview of food security and nutrition. Addressing the threat from climate variability and extremes for food and security and nutrition. Accra: FAO and ECA.

FAOSTAT (2023). FAOSTAT statistical database. [Rome]: FAO. Available at: https://www.fao.org/faostat/en/#data/FBS (Accesed May 6, 2023).

Farmery, A. K., Scott, J. M., Brewer, T. D., Eriksson, H., Steenbergen, D. J., Albert, J., et al. (2020). Aquatic foods and nutrition in the Pacific. *Nutrients* 12:3705. doi: 10.3390/nu12123705

Fayad, D. (2023). Food insecurity and climate shocks in Madagascar: Republic of Madagascar. Selected Issues Papers 2023:1. doi: 10.5089/9798400242601.018

Ferguson, C. E. (2021). A rising tide does not lift all boats: intersectional analysis reveals inequitable impacts of the seafood trade in fishing communities. *Front. Mar. Sci.* 8:625389. doi: 10.3389/fmars.2021.625389

Feyaerts, H., Van den Broeck, G., and Maertens, M. (2020). Global and local food value chains in Africa: a review. *Agric. Econ.* 51, 143–157. doi: 10.1111/agec.12546

Fiorella, K. J., Okronipa, H., Baker, K., and Heilpern, S. (2021). Contemporary aquaculture: implications for human nutrition. *Curr. Opin. Biotechnol.* 70, 83–90. doi: 10.1016/j.copbio.2020.11.014

Food and Agriculture Organization (FAO) (2017a). Guidelines for Sustainable Aquaculture. Available at: https://www.fao.org/in-action/gsa/resources/publications/fr/ (Accessed June 17, 2023).

Food and Agriculture Organization (FAO) (2017b). The future of food and agriculture: Trends and challenges 2017. Available at: https://www.fao.org/3/i6583e/i6583e.pdf (Accessed June 9, 2023).

Food and Agriculture Organization (FAO) (2018). The state of world fisheries and aquaculture 2018 – meeting the sustainable development goals. Rome: FAO.

Food and Agriculture Organization (FAO) (2021). Fishery and Aquaculture Statistics 2019/FAO annuaire. Statistiques des pêches et de l'aquaculture 2019/FAO anuario. Estadísticas de pesca y acuicultura 2019.

Food and Agriculture Organization (FAO) (2022a). In brief to the state of world fisheries and aquaculture 2022. Rome: FAO.

Food and Agriculture Organization (FAO) (2022b). The state of world fisheries and aquaculture 2022. Available at: https://www.fao.org/3/cc0461en/cc0461en.pdf (Accessed May 6, 2023).

Franz, N., Smith, H., and Westlund, L. (2019). Towards the implementation of the SSF Guidelines in West and Central Africa. FAO Fisheries and Aquaculture Proceedings (FAO) eng no. 62.

Gashu, D., Demment, M. W., and Stoecker, B. J. (2019). Challenges and opportunities to the African agriculture and food systems. *Afr. J. Food Agric. Nutr. Dev.* 19, 14190–14217. doi: 10.18697/ajfand.84.BLFB2000

George, G., Corbishley, C., Khayesi, J. N., Haas, M. R., and Tihanyi, L. (2016). Bringing Africa in: promising directions for management research. *Acad. Manag. J.* 59, 377–393. doi: 10.5465/amj.2016.4002

Gephart, J. A., Golden, C. D., Asche, F., Belton, B., Brugere, C., Froehlich, H. E., et al. (2020). Scenarios for global aquaculture and its role in human nutrition. *Rev. Fisher. Sci. Aquac.* 29, 122–138. doi: 10.1080/23308249.2020.1782342

Gephart, J. A., Henriksson, P. J., Parker, R. W., Shepon, A., Gorospe, K. D., Bergman, K., et al. (2021). Environmental performance of blue foods. *Nature* 597, 360–365. doi: 10.1038/s41586-021-03889-2

Gitz, V., Meybeck, A., Lipper, L., Young, C., and Braatz, S. (2016). *Climate Change and FoodSecurity: Risks and Responses*. Food and Agriculture Organization of the United Nations.

GlobeFish (2023). Fish trade in Africa: An update. Available at: https://www.fao.org/ in-action/globefish/fishery-information/resource-detail/en/c/338418/ (Accessed June 12, 2023).

Golden, C. D., Koehn, J. Z., Shepon, A., Passarelli, S., Free, C. M., Viana, D. F., et al. (2021). Aquatic foods to nourish nations. *Nature* 598, 315–320. doi: 10.1038/s41586-021-03917-1

Gomez-Zavaglia, A., Mejuto, J. C., and Simal-Gandara, J. (2020). Mitigation of emerging implications of climate change on food production systems. *Food Res. Int.* 134:109256. doi: 10.1016/j.foodres.2020.109256

Gormaz, J. G., Fry, J. P., Erazo, M., and Love, D. C. (2014). Public health perspectives on aquaculture. *Curr. Environ. Health Reports* 1, 227–238. doi: 10.1007/s40572-014-0018-8

Graham, R. E. (2023). The marine litter issue in the Windward Islands-a pathway to responses using the DPSIR framework. *Front. Environ. Sci.* 11:389. doi: 10.3389/ fenvs.2023.1150722

Guenard, R. (2021). Poisson from a petri dish. Inform 32:229. doi: 10.4060/ca9229en

Gutierrez, A. (2015). The sustainable seafood movement: Bringing together supply, demand, and governance of capture fisheries in the US and UK to achieve sustainability (Doctoral dissertation). Oxford: University of Oxford.

Gutiérrez, S., Svahn, S. L., and Johansson, M. E. (2019). Effects of omega-3 fatty acids on immune cells. *Int. J. Mol. Sci.* 20:5028. doi: 10.3390/ijms20205028

Haque, A. B., Cavanagh, R. D., and Seddon, N. (2021). Evaluating artisanal fishing of globally threatened sharks and rays in the bay of Bengal, Bangladesh. *PloS One* 16:e0256146. doi: 10.1371/journal.pone.0256146

Hinrichsen, E., Walakira, J. K., Langi, S., Ibrahim, N. A., Tarus, V., Badmus, O., et al. (2022). Prospects for Aquaculture Development in Africa: A review of past performance to assess future potential. *ZEF Working Papers*, Center for Development Research (ZEF), University of Bonn.

Hollinger, F., and Staatz, J. M. (2015). Agricultural growth in West Africa. Market and policy drivers. Federal Capital Territory: African Development Bank (ECOWAS).

Ikram, M., Rashid, M. T., and Shaukat, M. (2023). Climate change: A cataclysmic threat to humanity. *Journal of Xi'an Shiyou University, Natural Science Edition.* 9, 635–645.

Iversen, A., Asche, F., Hermansen, Ø., and Nystøyl, R. (2020). Production cost and competitiveness in major salmon farming countries 2003–2018. *Aquaculture* 522:735089. doi: 10.1016/j.aquaculture.2020.735089

Jennings, S., Stentiford, G. D., Leocadio, A. M., Jeffery, K. R., Metcalfe, J. D., Katsiadaki, I., et al. (2016). Aquatic food security: insights into challenges and solutions from an analysis of interactions between fisheries, aquaculture, food safety, human health, fish and human welfare, economy and environment. *Fish Fish*. 17, 893–938. doi: 10.1111/faf.12152

Jolly, C. M., Nyandat, B., Yang, Z., Ridler, N., Matias, F., Zhang, Z., et al. (2023). Dynamics of aquaculture governance. *J. World Aquacult. Soc.* 54, 427–481. doi: 10.1111/ jwas.12967

Kaleem, O., and Sabi, A. F. B. S. (2021). Overview of aquaculture systems in Egypt and Nigeria, prospects, potentials, and constraints. *Aquaculture Fisheries* 6, 535–547. doi: 10.1016/j.aaf.2020.07.017

Katikiro, R. E., and Macusi, E. D. (2012). Impacts of climate change on west African fisheries and its implications on food production. *J. Environ. Sci. Manag.* 15.

Khan, A., Ahmed, S. M., Sarr, C., Kabore, Y., Kahasha, G., Bangwe, L., et al. (2021). Nourishing nations during pandemics: why sprioritise fish diets and aquatic foods in Africa. *Maritime Studies* 20, 487–500. doi: 10.1007/s40152-021-00236-z Kolding, J., Zwieten, P. A., van Marttin, F., Funge-Smith, S., and Poulain, F. (2019). Freshwater small pelagic fish and fisheries in the main African great lakes and reservoirs in relation to food security and nutrition.

Kruk, S., and Peters, H. (2018). Towards a socially responsible aquaculture stewardship council. Oxford Briefing Notes, 1–18.

Kyvelou, S. S., Ierapetritis, D. G., and Chiotinis, M. (2023). The future of fisheries co-Management in the Context of the sustainable blue economy and the Green Deal: there is no Green without blue. *Sustain. For.* 15:7784. doi: 10.3390/su15107784

Latino, L. R., Pica-Ciamarra, U., and Wisser, D. (2020). Africa: The livestock revolution urbanizes. *Glob. Food Secur.* 26:100399.

Laso, J., Ruiz-Salmón, I., Margallo, M., Villanueva-Rey, P., Poceiro, L., Quinteiro, P., et al. (2022). Achieving sustainability of the seafood sector in the European Atlantic area by addressing eco-social challenges: the NEPTUNUS project. *Sustain. For.* 14:3054. doi: 10.3390/su14053054

Looy, H., Dunkel, F. V., and Wood, J. R. (2014). How then shall we eat? Insect-eating attitudes and sustainable foodways. *Agric. Hum. Values* 31, 131–141. doi: 10.1007/s10460-013-9450-x

Lubembe, S. I., Turyasingura, B., and Chavula, P. (2022). Reflection on impacts of climate change on fisheries and aquaculture: sub-Sahara Africa. *Weather* 6, 62–72.

Lucrezi, S. (2022). Public perceptions of marine environmental issues: a case study of coastal recreational users in Italy. *J. Coast. Conserv.* 26, 52–14. doi: 10.1007/s11852-022-00900-4

Mansour, A. T., Allam, B. W., Srour, T. M., Omar, E. A., Nour, A. A. M., and Khalil, H. S. (2021). The feasibility of monoculture and polyculture of striped catfish and Nile tilapia in different proportions and their effects on growth performance, productivity, and financial revenue. *J. Marine Sci. Eng.* 9:586. doi: 10.3390/jmse9060586

Mateo-Sagasta, J., Zadeh, S. M., Turral, H., and Burke, J. (2017). Water pollution from agriculture: a global review. Executive summary. Rome: FAO.

Maulu, S., Nawanzi, K., Abdel-Tawwab, M., and Khalil, H. S. (2021). Fish nutritional value as an approach to children's nutrition. *Front. Nutr.* 8:1090. doi: 10.3389/fnut.2021.780844

Mbaka, M., Cox, J. H., and Ronan, S. (2022). Aquaculture in Africa: Aquatic animal welfare, impact on the environment, and the sustainability of the sector. Available from the Wellbeing International Studies Repository at: https://www.wellbeingintlstudiesrepository.org/aw\_farm\_aqua/1. (Assessed November 3, 2023).

McClanahan, T. R., Maina, J. M., and Muthiga, N. A. (2011). Associations between climate stress and coral reef diversity in the western Indian Ocean. *Glob. Chang. Biol.* 17, 2023–2032. doi: 10.1111/j.1365-2486.2011.02395.x

McClure, S. B. (2015). The pastoral effect: niche construction, domestic animals, and the spread of farming in Europe. *Curr. Anthropol.* 56, 901–910. doi: 10.1086/684102

Meyer-Rochow, V. B. (2009). Food taboos: their origins and purposes. J. Ethnobiol. Ethnomed. 5, 1–10. doi: 10.1186/1746-4269-5-18

Mindjimba, K., Rosenthal, I., Diei Ouadi, Y., Bomfeh, K., and Randrianantoandro, A. (2019). FAO-Thiaroye processing technique: Towards adopting improved fish smoking systems in the context of benefits, trade-offs and policy implications from selected developing countries. Rome: FAO.

Mitra, S., Chakraborty, A. J., Tareq, A. M., Emran, T. B., Nainu, F., Khusro, A., et al. (2022). Impact of heavy metals on the environment and human health: novel therapeutic insights to counter the toxicity. *J. King Saud University Sci.* 34:101865. doi: 10.1016/j. iksus.2022.101865

Mittal, S., Gandhi, S., and Tripathi, G. (2010). Socio-economic impact of mobile phones on Indian agriculture (No. 246). Working paper.

Mpumi, N., Machunda, R. S., Mtei, K. M., and Ndakidemi, P. A. (2020). Selected insect pests of economic importance to *Brassica oleracea*, their control strategies and the potential threat to environmental pollution in Africa. *Sustain. For.* 12:3824. doi: 10.3390/ sul2093824

Muringai, R. T., Mafongoya, P., and Lottering, R. T. (2022). Sub-Saharan Africa freshwater fisheries under climate change: a review of impacts, adaptation, and mitigation measures. *Aust. Fish.* 7:131. doi: 10.3390/fishes7030131

Narayan, D., and Petesch, P. (2002). Voices of the poor: From many lands. Washington, DC: World Bank and Oxford University Press.

Nath, S. S., Bolte, J. P., Ross, L. G., and Aguilar-Manjarrez, J. (2000). Applications of geographical information systems (GIS) for spatial decision support in aquaculture. *Aquac. Eng.* 23, 233–278. doi: 10.1016/S0144-8609(00)00051-0

Naylor, R., Fang, S., and Fanzo, J. (2023). A global view of aquaculture policy. *Food Policy* 116:102422. doi: 10.1016/j.foodpol.2023.102422

Naylor, R. L., Kishore, A., Sumaila, U. R., Issifu, I., Hunter, B. P., Belton, B., et al. (2021). Blue food demand across geographic and temporal scales. *Nat. Commun.* 12:5413. doi: 10.1038/s41467-021-25516-4

NEPAD Planning and Coordinating Agency (2016). The Pan-African fisheries and aquaculture policy framework and reform strategy: Key policy messages transition to sustainability, profitability and equitable distribution, vol. 7. NEPAD Planning and Coordinating Agency.

Nguyen, N. H. (2016). Genetic improvement for important farmed aquaculture species with a reference to carp, tilapia and prawns in Asia: achievements, lessons and challenges. *Fish Fish.* 17, 483–506. doi: 10.1111/faf.12122

Nwozor, A., Olanrewaju, J., and Ake, M. (2019). National insecurity and the challenges of food security in Nigeria. *AJIS* 8, 9–20. doi: 10.36941/ajis-2019-0032

Obiero, K., Meulenbroek, P., Drexler, S., Dagne, A., Akoll, P., Odong, R., et al. (2019). The contribution of fish to food and nutrition security in eastern Africa: emerging trends and future outlooks. *Sustain. For.* 11:1636. doi: 10.3390/su11061636

Odeku, K. O., and Paulos, B. M. (2017). Prohibition of pollution of marine environments: challenges and prospects. *Environ. Econ.* 8, 127–136. doi: 10.21511/ ee.08(3-1).2017.05

Odende, T., Ogello, E. O., Iteba, J. O., Owori, H., Outa, N., Obiero, K. O., et al. (2022). Promoting sustainable smallholder aquaculture productivity through landscape and seascape aquapark models: a case study of Busia County, Kenya. *Front. Sustain. Food Syst.* 6:898044. doi: 10.3389/fsufs.2022.898044

Okocha, R. C., Olatoye, I. O., and Adedeji, O. B. (2018). Food safety impacts of antimicrobial use and their residues in aquaculture. *Public Health Rev.* 39, 21–22. doi: 10.1186/s40985-018-0099-2

Okon, E. M., Okocha, R. C., Adesina, B. T., Ehigie, J. O., Alabi, O. O., Bolanle, A. M., et al. (2022). Antimicrobial resistance in fish and poultry: public health implications for animal source food production in Nigeria, Egypt, and South Africa. *Front. Antibiot.* 1:1043302. doi: 10.3389/frabi.2022.1043302

Okyere, I., Chuku, E. O., Dzantor, S. A., Ahenkorah, V., and Adade, R. (2023). Capacity deficit and marginalization of artisanal fishers hamper effective fisheries governance in Ghana: insights and propositions for promoting sustainable small-scale fisheries. *Mar. Policy* 153:105640. doi: 10.1016/j.marpol.2023.105640

Ollivier, T., and Giraud, P. N. (2011). Assessing sustainability, a comprehensive wealth accounting prospect: an application to Mozambique. *Ecol. Econ.* 70, 503–512. doi: 10.1016/j.ecolecon.2010.09.028

Oreggioni, G. D., Ferraio, F. M., Crippa, M., Muntean, M., Schaaf, E., Guizzardi, D., et al. (2021). Climate change in a changing world: socio-economic and technological transitions, regulatory frameworks and trends on global greenhouse gas emissions from EDGAR v. 5.0. *Glob. Environ. Chang.* 70:102350. doi: 10.1016/j.gloenvcha.2021.102350

Popkin, B. M. (1994). The nutrition transition in low-income countries: an emerging crisis. *Nutr. Rev.* 52, 285–298. doi: 10.1111/j.1753-4887.1994.tb01460.x

Qadri, R., and Faiq, M. A. (2020). Freshwater pollution: effects on aquatic life and human health. *Fresh water Pollut. Dyn. Remediat.*, 15–26. doi: 10.1007/978-981-13-8277-2\_2

Qian, F., Riddle, M. C., Wylie-Rosett, J., and Hu, F. B. (2020). Red and processed meats and health risks: how strong is the evidence? *Diabetes Care* 43, 265–271. doi: 10.2337/ dci19-0063

Ragasa, C., and Loison, S. A. (2023). *Policy issues and options in aquatic food systems: Review of frameworks, tools, and studies.* IFPRI Discussion Paper 2161. Washington, DC: International Food Policy Research Institute (IFPRI).

Rahman, M. M. (2021). The rhetorical blue economy in Bangladesh: sAnalysing the inborn regulatory bottlenecks. *Acta Aquatica* 8, 66–73. doi: 10.29103/aa.v8i2.4775

Rajapaksha, L., Gunathilake, D. M. C. C., Pathirana, S. M., and Fernando, T. (2021). Reducing post-harvest losses in fruits and vegetables for ensuring food security—case of Sri Lanka. *MOJ Food Process Technols* 9, 7–16. doi: 10.15406/mojfpt.2021.09.00255

Reddy, M. M., Jennings, L., and Thomas, O. P. (2021). Marine biodiscovery in a changing world. *Prog. Chem. Org. Nat. Prod.* 116, 1–36. doi: 10.1007/978-3-030-80560-9\_1

Reid, A. J., Carlson, A. K., Creed, I. F., Eliason, E. J., Gell, P. A., Johnson, P. T., et al. (2019). Emerging threats and persistent conservation challenges for freshwater biodiversity. *Biol. Rev.* 94, 849–873. doi: 10.1111/brv.12480

Rice, J. C., and Garcia, S. M. (2011). Fisheries, food security, climate change, and biodiversity: characteristics of the sector and perspectives on emerging issues. *ICES J. Mar. Sci.* 68, 1343–1353. doi: 10.1093/icesjms/fsr041

Rocha, C. P., Cabral, H. N., Marques, J. C., and Gonçalves, A. M. (2022). A global overview of aquaculture food production with a focus on the activity's development in transitional systems—the case study of a south European country (Portugal). *J. Marine Sci. Eng.* 10:417. doi: 10.3390/jmse10030417

Rolle, R. S. (2006). Improving post-harvest management and marketing in the Asia-Pacific region: issues and challenges. *Post-harvest Manag. Fruit Veg. Asia-Pacific Region* 1, 23–31

Sanjay Panda, R. K. M., and Diwan, A. D. (2023). Omega fatty acids of fishes and human brain development: a study on naturopathy and homoeopathy. *J. Survey Fisheries Sci.* 10, 4204–4213.

Schönfeldt, H. C., and Hall, N. G. (2012). Dietary protein quality and malnutrition in Africa. Br. J. Nutr. 108, S69–S76. doi: 10.1017/S0007114512002553

Shahidi, F., and Ambigaipalan, P. (2018). Omega-3 polyunsaturated fatty acids and their health benefits. *Annu. Rev. Food Sci. Technol.* 9, 345–381. doi: 10.1146/annurev-food-111317-095850

Sohrabpour, V., Hellström, D., and Jahre, M. (2012). Packaging in developing countries: identifying supply chain needs. *J. Humanitarian Logistics Supply Chain Manag.* 2, 183–205. doi: 10.1108/20426741211260750

Spalding, M. D., Fish, L., and Wood, L. J. (2008). Toward representative protection of the world's coasts and oceans—progress, gaps, and opportunities. *Conserv. Lett.* 1, 217–226. doi: 10.1111/j.1755-263X.2008.00030.x

Srinivasan, U. T., Watson, R., and Sumaila, U. R. (2012). Global fisheries losses at the exclusive economic zone level, 1950 to present. *Mar. Policy* 36, 544–549. doi: 10.1016/j. marpol.2011.10.001

Sugri, I., Abubakari, M., Owusu, R. K., and Bidzakin, J. K. (2021). Post-harvest losses and mitigating technologies: evidence from upper east region of Ghana. *Sustainable Futures* 3:100048. doi: 10.1016/j.sftr.2021.100048

Tacon, A. G. (2023). Contribution of fish and seafood to global food and feed supply: an analysis of the FAO food balance sheet for 2019. *Rev. Fisheries Sci. Aquac.* 31, 274–283. doi: 10.1080/23308249.2022.2124364

Tacon, A. G., Lemos, D., and Metian, M. (2020). Fish for health: improved nutritional quality of cultured fish for human consumption. *Rev. Fish. Sci. Aquac.* 28, 449–458. doi: 10.1080/23308249.2020.1762163

Temudo, M. P., and Abrantes, M. (2014). The cashew frontier in Guinea-Bissau, West Africa: changing landscapes and livelihoods. *Hum. Ecol.* 42, 217–230. doi: 10.1007/s10745-014-9641-0

Thurstan, R. H., and Roberts, C. M. (2014). The past and future of fish consumption: can supplies meet healthy eating recommendations? *Mar. Pollut. Bull.* 89, 5–11. doi: 10.1016/j.marpolbul.2014.09.016

Tigchelaar, M., Leape, J., Micheli, F., Allison, E. H., Basurto, X., Bennett, A., et al. (2022). The vital roles of blue foods in the global food system. *Glob. Food Sec.* 33:100637. doi: 10.1016/j.gfs.2022.100637

Troell, M., Naylor, R. L., Metian, M., Beveridge, M., Tyedmers, P. H., Folke, C., et al. (2014). Does aquaculture add resilience to the global food system? *Proc. Natl. Acad. Sci.* 111, 13257–13263. doi: 10.1073/pnas.1404067111

Uddin, M. M., Schneider, P., Asif, M. R. I., Rahman, M. S., and Mozumder, M. M. H. (2021). Fishery-based ecotourism in developing countries can enhance the social-ecological resilience of coastal fishers—a case study of Bangladesh. *Water*, 13:292.

Venugopal, V., and Gopakumar, K. (2017). Shellfish: nutritive value, health benefits, and consumer safety. *Compr. Rev. Food Sci. Food Saf.* 16, 1219–1242. doi: 10.1111/1541-4337.12312

Vermunt, D. A., Wojtynia, N., Hekkert, M. P., Van Dijk, J., Verburg, R., Verweij, P. A., et al. (2022). Five mechanisms blocking the transition towards 'nature-inclusive'agriculture: a systemic analysis of Dutch dairy farming. *Agric. Syst.* 195:103280. doi: 10.1016/j.agsy.2021.103280

Wilson, D. C., and McCay, B. J. (2019). Fishery management, human dimension. *Encyclopedia of Ocean Sciences*, volume 2, pp 1023–1018, Elsevier Ltd.

Wood, B., Williams, O., Nagarajan, V., and Sacks, G. (2021). Market strategies used by processed food manufacturers to increase and consolidate their power: a systematic review and document analysis. *Glob. Health* 17, 1–23. doi: 10.1186/s12992-021-00667-7

Worldometer (2023). Africa population. Available at: https://www.worldometers.info/ world-population/africa-population/ (Accessed June 17, 2023).

Yitayew, M., and Melesse, A. M. (2011). Critical water resources issues in the Nile River basin. *Nile River Basin*, 401–416. doi: 10.1007/978-94-007-0689-7\_20

Zeller, D., Vianna, G. M., Ansell, M., Coulter, A., Derrick, B., Greer, K., et al. (2021). Fishing effort and associated catch per unit effort for small-scale fisheries in the Mozambique Channel region: 1950–2016. *Front. Mar. Sci.* 8:707999. doi: 10.3389/ fmars.2021.707999

Zhang, K., Zhang, F., Zeng, Z., and Ma, S. (2023). Impact of extreme climate events on marine ecosystems: adaptation and challenges. *Front. Mar. Sci.* 10:1187357. doi: 10.3389/fmars.2023.1187357

Zhao, W., Yin, C., Hua, T., Meadows, M. E., Li, Y., Liu, Y., et al. (2022). Achieving the sustainable development goals in the post-pandemic era. *Human. Soc. Sci. Commun.* 9, 258–257. doi: 10.1057/s41599-022-01283-5

Zilhão, J., Angelucci, D. E., Igreja, M. A., Arnold, L. J., Badal, E., Callapez, P., et al. (2020). Last interglacial Iberian Neandertals as fisher-hunter-gatherers. *Science* 367:eaaz7943